Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
March 16, 2011
Injector    Energy 61.76 MeV
· (3/11) The injector optics was optimized. The injector was set up for high current run (150uA). 
· Since Hall C has been down and Hall A only took low current beam. The injector needs fine tuning for high current before Hall C is back online. Reza is scheduled to do it sometime today. The 0L07 spectrometer is on again.
· The injector was good last week delivering beam to Hall A.

Accelerator   Linac Energy 549.00 MeV
· The accelerator did not deliver much beam last week due to the activities in the halls. The time was utilized for maintenance and improving the machine optics. Over the weekend we delivered low current to Halls A and C. 
· (3/15) Beam delivery was suspended for about 12 minutes to conduct a tornado drill. The drill went well. After the drill, four hours beam studies: Commissioning Helicity Magnets at 1kHz Helicity Reversal Rate. 
· (3/15) Four cavities in 1L20 were recovered.  
· (3/12) Zone 1L20 was bypassed. 

· Separators for pass 1 and pass 4 are on and beam was steered to the dumplettes of Hall C and Hall A. A is pass 4 and C pass 1. 2T line aperture is tight horizontally and we are going to address it today.
· (3/8) Maintenance day. 0L07 spectrometer on; 6E02 viewer replaced; Arc9A earth leakage sensor replaced; PSS maintenance and many other things (48 items were scheduled. See entry # 1573258 for details.).
· Last week some multiharp data were collected to improve QWeak beam quality.

Hall A---Pass 4, E08 - 010
· (3/15) Q1 is fixed.

· The hall did not take beam until Sunday owl shift (3/13). The hall had problems with its left Q1 magnet. The magnet would not ramp up to the desired current. 
· (3/9) 4-pass separation was done during owl shift. The hall will not be ready for swing shift. The hall line has not been setup yet.
· (3/8) The finished the current run. Switching to experiment E07-006. The vacuum in the Compton region has been degrading. The hall does not need the Compton.
Hall B ---Pass 5, TPE
· The hall has been down since Friday preparing a new experiment (HDice).
Hall C--- Pass 1, QWeak 
· The hall started to take beam Friday swing shift (3/11). It has been taking low current (< 25uA). The beam was fine. The BLA was calibrated at 70uA and needs to be calibrated at higher current (150uA). The hall will be down today to cool down its target.
· (3/15) MQE3M03 power supply would not turn on. Hall A is running with both Moeller quads off. EES is helping troubleshoot the problem. Hall C beam size will be large until the Moller quads is fixed.
· The hall is still working on its vacuum and will not be ready for beam until sometime tomorrow.  
